

EO-1 Weekly Report


	Earth Observing – 1

	February 15, 2012 – February 21, 2012

	DOY
	046
	–
	052

	Mission Day
	4164
	–
	4170


EO-1 SPACECRAFT

Instruments

· Scheduled 124 science Data Collection Events (DCEs) this week 

· Received all images
· Performed Instrument calibration

· ALI Internal Calibration II at 051/00:00z
EO-1 MISSION OPERATIONS CENTER

Real-Time

· Maintained nominal spacecraft state of health
· Leap Second Monitoring

· Step 1 – Change the script to remove the leap second after the GPS satellites begin broadcasting
· Complete

· Step 2 – At the time of the leap second upload the correct number of leap seconds

· Step 3 – Revert back to the previous script after the leap second is accounted for

· Monitoring the battery differential voltage, temperature, current, voltage and state of charge

· Four days after the lunar calibration (with the VT level still at 4.0) the EOD SOC was back to 99.5% or higher

· Planning on keeping the VT Level at 4.0 when not performing a lunar calibration

· Working with Flight Software people for EO-1 to modify the TSM limits

· Flight Software people are creating a load to change the TSM limits since we have seemed to plateau on the differential voltage

· The TRMM spacecraft has the same type of battery, 50 AH Super NiCad, and routinely sees differential voltages of +/-500mV

· Redesign of Lunar Calibration
· Despite the power reduction of removal of the 5th scan, the EO-1 Battery still reached 25.6 V, only .1 V over the TSM that would put the spacecraft into safe-hold so a redesign of the Lunar Calibration is beginning
· The redesigned Lunar Calibration will have the following profile

· Set VT level to 4.5

· One S/C Day cycle

· 3 scan phase 1 of lunar calibration

· One/Two S/C Day cycle(s)

· 3 scan phase 2 of lunar calibration

· Two S/C Day cycles for power return to 99.5% or more end of day state of charge

· Set VT level to 4.0

· Working with the mission scientists that use the lunar calibration to determine which scans would be on each phase

· TrollSat (TR2) is now certified for S-Band passes and will be putting 2-3 per week in the schedule
· In the coming weeks we will be setting up X-band tests with TR2

· Continuing to monitor SGS passes to ensure they remain stable
· SGS has an equipment problem and will be down until approximately August 2012
· SGS passes being taken on the SG-1/SG-22 hybrid antenna that was used during last equipment malfunction

Mission Planning
· Redesigning the Lunar Calibration to be two separate 3-scan phases
· Looking at the steps required in creating a South Pole image

· Remaking the SOP for mission planning activities for ASPEN and ASIST SCP
Flight Dynamics
· Step 1 of the Leap Second is complete, 
· Worked with FDF for getting products from their new Product Servers

· Finalizing certificate based SFTP with new product server

· Looking at the steps required to take a South Pole image
System Administration
· Completed scripts to simplify the process for viewing firewall logs. These scripts use the tags established in the firewall ruleset to narrow down the logs for quicker analysis.

· Configured certificate based SSH/SFTP from 14 operational systems to the backup server to improve the backup process.

· Created the backup script’s logging directory on three operational systems. This directory must not have been created when the backup script was initially implemented.

· Completed applying the Feb 2012 updates/patches on the IONet Windows computers. The patches included Java, Firefox, Adobe Flash (IE and FF), and Microsoft products.

· Completed applying the Feb 2012 updates/patches on the 6 CNE computers. The patches included Java, Firefox, Adobe Flash (IE and FF), and Microsoft products.

· Replaced failed monitor on the primary ASIST T&C system. The backup ASIST T&C system was reconfigured for single monitor use so that the primary could continue with dual screens.

· Corrected persistent paper jams associated with the MOC CNE printer. Replacing the transfer kit, rollers, and air filters appears to have returned the printer to a nominal state. 

· Provided IT Security group with the information they needed to update EO-1 documentation after the MOC relocation.

GROUND AND SPACE NETWORK

Station Downtimes 

· SGS has had an equipment malfunction and latest indication is that it will be back up next August
· SG1/22 has a communication line problem, workaround is in place, fix timing unknown
· HGS is down, unknown time to be back up

Operational Discrepancies

· None
Number of Real-Time Supports

[image: image1.emf]Date DOY S/X-Band S-Band Only Non-DSMC TDRSS Daily Totals

15-Feb-12 46

4 6 1 0

11

16-Feb-12 47

6 3 0 0

9

17-Feb-12 48

6 3 1 0
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18-Feb-12 49

6 3 0 0
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19-Feb-12 50

4 7 1 0

12

20-Feb-12 51

5 4 1 0
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21-Feb-12 52

4 4 0 0
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35 30 4 0 69 Weekly Totals


UPCOMING EVENTS

· Redesign of Lunar Calibration
· TSM update onboard spacecraft
· TR2 X-Band pass testing
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		Date		DOY		S/X-Band		S-Band Only		Non-DSMC		TDRSS		Daily Totals

		15-Feb-12		46		4		6		1		0		11

		16-Feb-12		47		6		3		0		0		9

		17-Feb-12		48		6		3		1		0		10

		18-Feb-12		49		6		3		0		0		9

		19-Feb-12		50		4		7		1		0		12

		20-Feb-12		51		5		4		1		0		10

		21-Feb-12		52		4		4		0		0		8

		Weekly Totals				35		30		4		0		69
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